Author's response to reviews

Title: Seroma suppression in mastectomy patients - a case report using TissuGlu

Authors:

Christian Eichler (ceichler@gmail.com)
Faten Dahdouh (dahdouhf@kliniken-koeln.de)
Axel Sauerwald (axel.sauerwald@gmx.de)
Mathias Warm (warmf@kliniken-koeln.de)

Version: 2 Date: 6 December 2012

Author's response to reviews:

The revisions were made as requested in your last email:

1. Please include the study design in your title, i.e. Case report. For example: A presenting with B in C: a case report.

2. Please restructure the Abstract into the following three sections: Introduction, Case presentation, and Conclusion. The abstract should be no longer than 350 words.

3. Please replace the header “Patient” to “Case Presentation.”

4. Please replace dates found in the case presentation to the amount of time this occurred before the presentation of the case report.

5. Please include the ethnicity of the patient in the case presentation section of the manuscript.

6. Please include a Consent section at the end of the manuscript, before the reference list. This should contain a statement to confirm that informed written consent was received for publication of the manuscript and figures. We recommend the following wording:

“Written informed consent was obtained from the patient for publication of this manuscript and accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal."

It is a requirement to have written informed consent from the patient for publication of the case and we will not proceed with your manuscript until you have confirmed this.
7. Please include a competing interests section at the end of the manuscript, before the reference list. If the authors have no competing interests, please state: "The authors declare that they have no competing interests."

8. Please include the authors contribution section. We suggest the following kind of format (please use initials to refer to each author’s contribution): FC analyzed and interpreted the patient data regarding the hematological disease and the transplant. RH performed the histological examination of the kidney, and was a major contributor in writing the manuscript. All authors read and approved the final manuscript.